Author's response to reviews

Title: Comparison between observed children's tooth brushing habits and those reported by mothers

Authors:

  Carolina C Martins (carolcm10@hotmail.com)
  Maria J Oliveira (lagesdeoliveira@gmail.com)
  Isabela A Pordeus (isabela@netuno.lcc.ufmg.br)
  Saul M Paiva (smpaiva@uol.com.br)

Version: 5 Date: 30 August 2011

Author's response to reviews: see over
On behalf of the other authors, I am resubmitting a revised version of the manuscript “Comparison between observed children’s tooth brushing habits and those reported by mothers” (MS: 5845938625136848) for the editorial analysis of BMC Oral Health.

Our responses to the topics addressed are presented below:

1. We removed all the effects of the Table and we let the highlights in bold.
2. We removed all the highlights of the text.
3. We removed the vertical lines of the Table.
4. I am sorry, but we could not understand what you mean with “merged cells where the merged cell crosses rows”. Could you please solve this problem for us?
5. We checked the text and we found no mistakes.

Yours sincerely,

Saul Martins Paiva